Author’s response to reviews

Title: A Prospective Study of Salvational Intervention with ICS/LABA for Reducing Chronic Obstructive Pulmonary Disease Exacerbation Under Severe Air Pollution (SIRCAP) in Beijing: protocol of a multi-center randomized controlled trial

Authors:

Tianyu Zhou (zhoutianyu@bjmu.edu.cn)
Yijue Zhong (zhongyijue2016@bjmu.edu.cn)
Jiping Liao (colorfulwing01@163.com)
Guangfa Wang (wangguangfa@hotmail.com)
Xueying Li (xyinglee@163.com)
Xiaosen Qian (qian186@sohu.com)
Pingchao Xiang (xpcemail@163.com)
Xinsheng Chen (Syhx2011@163.com)
Zhenyang Xu (azhenyang@sina.com)
Fengzhen Zhang (zfz731hx@163.com)
Xiaoguang Wang (wxg731@163.com)
Senlin Wang (13701387739@163.com)
Xiangxin Li (13691008082@163.com)
Yongxiang Zhang (zyx8669@163.com)
Guoguang Xia (guoguangxia1958@sina.com)
Li Dai (dailimd@hotmail.com)

Version: 1 Date: 25 Feb 2018

Author’s response to reviews:

1. Ethical and Funding Approval Documentation.

We have forwarded the requested documentation to BMCSeriesEditorial@biomedcentral.com.
2. Funding.

The funding of the study are both governmental funding and the study protocol has undergone peer-review by the funding body.

3. Study status

The study is currently at the stage of patient recruitment and data collection.

4. Related Articles

Until now not any result of this study has already been published or been submitted to any journal.

5. Tables

We have corrected the labels of tables.